Author’s response to reviews

Title: Comparative efficacy of placebos in short-term antidepressant trials for major depression: A secondary meta-analysis of placebo-controlled trials

Authors:
Lisa Holper (lisa.holper@bli.uzh.ch)
Michael P. Hengartner (heng@zhaw.ch)

Version: 1 Date: 30 Jun 2020

Author’s response to reviews:

Dear Editor,

Thank you for your message and details on formatting.

- The sentence "All authors have read and approved the manuscript" has been added to the contribution section.

Kind regards
Lisa Holper